18.099B section exercise on Basic LaTeX

Miki Havlíčková

Thursday 02/12

1 Introduction

This is a 1.2 spaced document, containing footnotes^1 and other basic LaTeX commands. If you can reproduce it already, you may skip next week’s section. If you are not familiar with LaTeX, or need a bit of brushing up on it, come on Tuesday 2/17 at 2:30 pm to room 2-136 for a session with Miki (note: that Tuesday is on Monday schedule).

Either way, your assignment is to reproduce this article and bring printouts of both the .dvi and .tex file to your section on 2/23 or 2/24. Miki’s tex file will be posted on the class website on Tuesday 2/24.

2 Table

Breakfast Lunch Dinner Monday Cereal Sandwich Chicken Wednesday Muffin Soup Pasta Friday Cereal Sandwich Stir-fry

3 Descriptions

Here are sample descriptions of the entries, in the appropriate list environment:

Cereal Favorite breakfast food

Muffin Bread-like pastry

Pasta Italian favorite

1like this one